CS 451 Programming Paradigms, Spring 2001 1

Homework 0 — due in class on Monday 22 January

Student Background Information 1. Courses taken at UNM or, Name & place of equivalent Semester/ Grade (optional) non-UNM course Year CS 201 Mathematical Foundations of C.S. EECE 238L Computer Logic Design CS 251L Intermediate Programming CS 257L Nonimperative Programming CS 341L Introduction to Computing Systems CS 351L Design of Large Programs CS 361L Data Structures and Algorithms I CS 441 Modern Computer Architecture CS 448 Design of Computers CS 454 Compiler Construction CS 484 Unix Administration and Tools 2. In which high-level programming languages have you successfully written programs, and especially, large projects?

Name Student ID Undergraduate major / Graduate dept. Undergraduate class / Graduate year of study Email